



BIS Working Papers No 1050

Population aging and bank risk-taking

by Sebastian Doerr, Gazi Kabaş and Steven Ongena

Monetary and Economic Department

November 2022

JEL classification: E51, G21.

Keywords: Risk-taking, financial stability, low interest rates, population aging, demographics.

BIS Working Papers are written by members of the Monetary and Economic Department of the Bank for International Settlements, and from time to time by other economists, and are published by the Bank. The papers are on subjects of topical interest and are technical in character. The views expressed in them are those of their authors and not necessarily the views of the BIS.

This publication is available on the BIS website (www.bis.org).

© Bank for International Settlements 2022. All rights reserved. Brief excerpts may be reproduced or translated provided the source is stated.

ISSN 1020-0959 (print) ISSN 1682-7678 (online)

Population aging and bank risk-taking

Sebastian Doerr

Gazi Kabaş

BIS

Tilburg

Steven Ongena UZH, SFI, KU Leuven & CEPR

November 2022

Abstract

What are the implications of an aging population for financial stability? To examine this question, we exploit geographic variation in aging across U.S. counties. We establish that banks with higher exposure to aging counties increase loan-toincome ratios, especially where they operate no branches. Laxer lending standards also lead to higher nonperforming loans during downturns, suggesting higher credit risk. Inspecting the mechanism shows that aging drives risk-taking through two contemporaneous channels: deposit inflows due to seniors' propensity to save in deposits; and depressed local investment opportunities due to seniors' lower credit demand. Banks thus look for riskier clients in no-branch counties.

JEL classification: E51, G21.

Keywords: Risk-taking, financial stability, low interest rates, population aging, demographics.

Sebastian Doerr is at the Bank for International Settlements (sebastian.doerr@bis.org). Gazi Kabaş is at Tilburg University (g.kabas@tilburguniversity.edu). Steven Ongena is at University of Zurich, Swiss Finance Institute, KU Leuven, NTNU, and CEPR (steven.ongena@bf.uzh.ch). We thank our discussants Bo Becker, Robin Greenwood, Andrew Mackinlay, George Pennacchi, Romina Ruprecht, Neeltje van Horen, and Jinyuan Zhang; as well as Adrien Auclert, Andreas Barth, Martin Brown, Tabea Bucher-Koenen, Andreas Fuster, Emilia Garcia-Appendini, Jose Jorge, Oguzhan Karakas, Luca Mazzone, Kasper Roszbach, and Simon Rother. We also thank participants at the American Finance Association Annual Conference, Financial Intermediation Research Society Conference, Western Finance Association 2022 Meeting, FDIC 20th Annual Bank Research Conference, JFI-Nova SBE Conference on Financial Intermediation and Corporate Finance, Norges Bank-CEPR Workshop on Frontier Research in Banking, European Economic Association Annual Conference, ZEW conference on Ageing and Financial Markets, SFI Research Days, Annual Meeting of the Swiss Society for Financial Market Research, and the Seminar for Contract Theory, Banking and Money at University of Zurich. Kabaş and Ongena gratefully acknowledge financial support from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme ERC ADG 2016 (No. 740272: lending). The views expressed here are those of the authors only, and not necessarily those of the Bank for International Settlements.

1 Introduction

The U.S. population of age 65 and above will grow by 18 million (or 33%) over the next decade, and similar developments are taking place in most other advanced economies. This unprecedented rise in the senior population has led to a debate on whether population aging leads to a build-up of risks in the financial sector. High savings rates among seniors might lead to an abundance of savings and depressed returns, thereby potentially encouraging banks to reach for yield (IMF, 2019).¹ Low expected returns and banks' reach for yield, in turn, could give rise to financial instabilities (Brookings, 2019).

This paper provides the first evidence on how an aging population affects bank lending standards. For identification, we exploit the extensive variation in aging that has already occurred across U.S. counties, combined with granular data on bank mortgage loans and deposits. We find that banks more exposed to aging counties relax lending standards as they grant new loans: loan-to-income ratios increase, and in particular in counties where banks operate no branches. Banks with greater exposure also see a sharper rise in nonperforming loans during downturns, implying an increase in credit risk due to population aging. Further analysis suggests that these patterns are shaped by two forces: an increase in banks' available funds due to seniors' higher propensity to save in the form of deposits, and a contemporaneous aging-induced decrease in the local demand for credit.

To measure banks' presence in aging counties and how it affects lending standards, we define bank exposure as the weighted average change in the number of seniors across counties where banks have branches. Weights are given by deposit shares at the beginning of the sample period. Intuitively, banks with higher exposure have a larger footprint in counties that see a stronger increase in seniors. Our main analysis focuses on the 1997–2007 period, and does so for three reasons. First, the period after the financial crisis of 2007/08 is characterized by substantial financial regulatory reforms. These reforms have affected banks' lending decisions and hamper an identification of the respective channels through which population aging affects bank risk-taking. Second, we can exploit the Great Recession as a negative shock to see whether aging-induced risk-taking manifests itself in higher nonperforming loans during a downturn. And third, we avoid the zero lower bound on interest rates (Leahy et al., 2022).

¹Indeed, population aging is a major driver of falling interest rates (Carvalho et al., 2017; Conesa and Kehoe, 2018), and low rates have been associated with higher risk-taking (Dell'Ariccia et al., 2017; Heider et al., 2021).

Bank exposure could be correlated with observable or unobservable bank characteristics, posing a challenge to identification. We address this challenge in a number of ways. First, we document that high- and low-exposure banks are similar in terms of initial balance sheet characteristics, for example in terms of size, capital ratio, or return on assets. In line with the balancedness in bank co-variates, in regressions we find that including bank controls and bank fixed effects barely affects the magnitude of our estimated coefficients, despite increasing the R^2 substantially. Bank exposure to aging counties is hence likely uncorrelated with observable and unobservable bank characteristics, reducing potential concerns about self-selection and omitted variable bias (Altonji et al., 2005; Oster, 2019).

Second, we develop an instrumental variable approach that exploits the predetermined component of counties' age structure. Specifically, we predict the change in the population of age 65 and above from 1997 to 2007 with the change in the population of age 45 to 65 from 1977 to 1987 in the same county. We then re-construct exposure, but based on aging predicted from historical trends, and use it as an instrumental variable (IV) for actual bank exposure to aging counties. This approach builds on the assumption that the historical demographic structure is plausibly exogenous to *changes* in contemporaneous confounding factors. Such factors include, for example, changes in life expectancy or economic conditions. To support the plausible exogeneity of our instrument, we show that directly controlling for banks' exposure to contemporaneous changes in county characteristics (e.g. changes in income per capita or the unemployment rate) does not affect our conclusions.

Armed with our measure of bank exposure, we establish that banks more exposed to aging counties relax their lending standards. As the main measure for risk-taking we focus on the loan-to-income (LTI) ratio, which is highly correlated with ex-post default (Fuster et al., 2021). Our results show that a 33 percentage point (pp) increase in bank exposure is associated with a significant rise in LTI ratios by around 23 pp.² The effect of exposure on LTI ratios is particularly strong in counties where banks have no branches. As a greater borrower distance to the nearest branch can result in less efficient screening and monitoring (DeYoung et al., 2008; Granja et al., 2018; Liberti and Petersen, 2019),

²Note that the increase of 33% in the number of seniors corresponds to the expected U.S. growth in the senior population over the next decade, from 2021 to 2030. The unconditional mean and standard deviation of the county-level growth in the number of seniors between 1997 and 2007 equal 12% and 15%, respectively. An increase in bank exposure by 33 pp occurs hence, for example, when moving from a bank with zero exposure to a bank with 33% exposure, which is a bank exposed to counties that deposit-weighted have a growth in seniors equal to the expected growth in the U.S. over the next decade.

the rise in LTI ratios in no-branch counties suggests an increase in credit risk.³ Local aging could hence not only lower lending standards in general, but also have geographic spillover effects to financial stability in *other* markets.

The rise in LTI ratios could in principle be explained by differences in the type of markets that more and less exposed banks serve. To absorb all observable and unobservable borrower-county characteristics, including loan demand, we exploit the granularity of our bank-county level data to include borrower-county fixed effects. Intuitively, these fixed effects enable us to compare two banks with different exposures that lend to the same county, mitigating concerns that risk-taking is driven by changes in borrower-county fundamentals (Khwaja and Mian, 2008; Jiménez et al., 2014). We still find a statistically and economically significant effect of exposure on lending standards when we enrich our specifications with borrower-county fixed effects.

The relaxation of lending standards among exposed banks suggests an increase in credit risk, with potentially negative consequences for financial stability. To investigate this aspect in more detail, we show that banks with higher exposure to aging counties see a significantly stronger increase in their share of nonperforming loans during the Great Recession. Importantly, controlling for banks' exposure to the rise in house prices prior to the Great Recession does not affect our estimates. Exposure to aging counties had effects on risk-taking above and beyond banks' exposure to the housing boom. Moreover, our findings hence suggest that the aging-induced relaxation in lending standards negatively affects financial stability during downturns.

Having established a strong association between population aging and bank risktaking, we investigate the underlying channels that could explain the link. Population aging could affect banks through changes in the supply of and demand for capital. On the one hand, Becker (2007) argues that seniors are more likely to save in the form of deposits, which are a stable and cheap source of funding (Hanson et al., 2015; Carletti et al., 2021). Higher exposure to aging counties could hence lead to deposit inflows and reduce banks' cost of funds, translating into an increase in lending but allowing banks to pursue safer projects with lower returns. On the other hand, an aging population could reduce the local demand for credit and depress returns. Homeownership rates and savings

³Banks could have opened branches between 1997 and 2007 in counties where they had no branch in 1997, possibly mitigating informational frictions in the lending process, leading to a smaller increase in LTI ratios. Consistent with this interpretation, we find that estimated coefficients increase in magnitude when we focus on counties where banks had no branches in 1997 and did not open any branches during the sample period.

are highest among seniors, and they are less likely to start new companies (Azoulay et al., 2020). An aging-induced decline in the labor force also reduces firms' marginal product of capital, further lowering the demand for capital (Auclert et al., 2021). Aging-exposed banks could hence be confronted with a decline in the demand for credit in counties where they have branches at a time when they experience an increase in available funds. Banks might therefore have to look for new and potentially riskier clients, in particular in counties where they operate no branches.

We first investigate the link between aging and deposits. We find that an increase in the number of seniors in a county has a positive and strongly significant effect on local bank deposits. A rise in a county's senior population by one-third is associated with an increase in county-level bank deposits by around 25%. Seniors' relatively larger share of wealth held in the form of deposits explains this positive relationship (Becker, 2007). For identification, our regressions absorb all observable and unobservable bank heterogeneity with bank fixed effects. They also include a large set of county-level control variables. We hence account for other bank or observable county characteristics that could explain the rise in deposits. In addition, we show that neither local aging nor bank exposure predict where banks open new branches prior to our sample period. This mitigates the concern that banks strategically opened branches to benefit from future deposit inflows in anticipation of local demographic trends.

We then establish that population aging dampens the local demand for credit. First, we use data from the Survey of Consumer Finances and show that seniors are less likely to borrow or have outstanding debt compared to younger respondents. These correlations hold even after controlling for an extensive set of household-level controls. Second, we decompose mortgage loan growth in each county into supply and demand factors, following Amiti and Weinstein (2018). Higher growth among seniors has a significant negative effect on the demand factor, suggesting that aging leads to a decline in the local demand for mortgages.

The aging-induced deposit inflows and weaker local demand for credit affect the geography of bank lending. Higher exposure leads banks to increase their mortgage lending. The increase in lending is especially large in counties where banks operate no branches, i.e., in counties where exposed banks are not directly affected by an aging-induced decline in credit demand. These results speak directly to our finding that banks relax their lending standards by most in counties where they operate no branches. They also add to concerns that banks exposed to aging markets may reach for yield outside their core markets (IMF, 2019).

To examine the robustness of our findings, we perform a series of additional tests. We start by investigating how the effect of aging on risk-taking changes along the LTI distribution. We find that the effect is stronger in the right tail of the LTI ratio distribution, suggesting higher risk-taking among borrowers that already had high LTI ratios. We also show that the risk-taking in no-branch counties is stronger in the right tail of the LTI ratio distribution. In line with the large literature documenting that banks with lower capital ratios engage more in risk-taking, we further find that higher exposure leads to a significantly higher increase in LTI ratios for low-capitalized banks. Next, we consider the share of denied loans as another measure of bank lending standards. Similar to our main results, we find that exposure to aging counties reduces the share of denied loans and that the effect is stronger in no-branch counties.

Finally, we also investigate whether laxer lending standards in the form of loan-toincome ratios affect household leverage. The results show that counties in which exposed banks have a larger ex-ante market share also see a stronger increase in household debtto-income ratios over the sample period. That is, the increase in risk-taking and *loan*-toincome ratios at the bank-county level is mirrored in an increase of households' *debt*-toincome ratios at the county level.

To the best of our knowledge, our paper is the first to investigate the effects of population aging on lending standards, and to what extent there are spillovers across markets.⁴ While this paper's focus is on risk-taking, our paper builds on work that shows how changes in banks' deposit base affect their loan volumes. Becker (2007) uses the local demographic structure as an instrument for changes in bank deposits, but focuses on how the availability of deposits affects local entrepreneurial activity. Gilje et al. (2016) provide evidence that banks exposed to local liquidity inflows from oil and natural gas shale discoveries subsequently increase lending in other counties where they have branches. They find that the effects are especially pronounced for loan types that are subject to more contracting frictions.⁵ Our novel results for population aging show that local aging not

⁴Recent papers investigate the effects of population aging on various outcomes. Butler and Yi (2021) investigate the effect of an aging population on the U.S. municipal bond market. Kopecky and Taylor (2022) use historical panel data to show that demographic shifts are correlated with asset returns and risk premia. Cravino et al. (2022) show that US population aging accounted for about 20% of the increase in the service share in consumption.

⁵For further papers on the effects of local shocks on bank lending, see Cortés and Strahan (2017); Smolyansky (2019); De Jonghe et al. (2020); Doerr and Schaz (2021) and Rehbein and Ongena (2021), among many others. In addition, Kundu et al. (2022) demonstrate that the geographic concentration of

only leads to deposit inflows, but also reduces the local demand for credit. The combination of both forces explains why banks increase their risk-taking, and why there are spillover effects to other no-branch markets.

We also contribute to the literature that explores how banks adjust their lending standards during credit booms (Berger and Udell, 2004; Dell'Ariccia and Marquez, 2006; Mendoza and Terrones, 2008; Dell'Ariccia et al., 2012; Justiniano et al., 2019) and periods of low interest rates (Maddaloni and Peydró, 2011; Altunbas et al., 2014; Jiménez et al., 2014; Ioannidou et al., 2015; Dell'Ariccia et al., 2017; Heider et al., 2021). These papers show that credit booms or periods of low interest rates can lead to laxer lending standards and a higher risk of financial crises. However, evidence on the effects of population aging on bank risk-taking is scarce, despite the significant policy attention devoted to this major macroeconomic trend (CGFS, 2018; ECB, 2018; IMF, 2019; OECD, 2019). Our analysis aims to fill this gap in the literature. As advanced economies face an unprecedented increase in the number of seniors over the next decade, the bank risk-taking channel of population aging could gain in importance for financial stability.⁶

The rest of the paper proceeds as follows. Section 2 describes our main data sources and the construction of the main variables. Section 3 explains the empirical strategy and presents our main results on risk-taking and financial stability. Section 4 then investigates the underlying mechanism. Section 5 provides additional tests, and Section 6 concludes.

2 Data and descriptive statistics

This section explains the construction of our main variables and reports descriptive statistics. Table OA1 in the Online Appendix provides further details on variable definitions and sources.

Our main analysis focuses on the period from 1997 to 2007, for three reasons. First, while the growth in the senior population has been even more pronounced after the Great Financial Crisis, the post-crisis period is characterized by substantial financial regulatory reform (such as the Dodd-Frank Act and regular stress tests) and encompassing government programs. These reforms have shaped banks' lending decisions, particularly mortgage lending, over the post-crisis period, which would make a clean identification of

deposits affects how shocks are transmitted across geographies by multi-market banks.

 $^{^{6}}$ We thereby also complement results in Drechsler et al. (2021), Lin (2020) and Doerr et al. (2021) by highlighting an alternative channel through which macroeconomic changes affect the financial sector.

the respective channels through which population aging affects bank risk-taking difficult. The absence of major financial regulatory changes during our sample period makes it well-suited to identify the effects of population aging on bank risk-taking. Second, we can exploit the Great Recession to analyze whether higher risk-taking in the years leading up to the crisis manifested itself in higher nonperforming loans during the shock episode. That is, we can analyze whether laxer lending standards had an impact on financial stability. And third, we avoid the zero lower bound on interest rates (Leahy et al., 2022).

2.1 Main variables

Population aging. Our main explanatory variable at the county level is the change in the log of the population of ages 65 and above from 1997 to 2007, denoted by Δold_c . We use the change in the number of seniors (i.e., in the level) rather than the change in the ratio of seniors to the total population (i.e., in the share), because changes in the share could be driven by changes in the numerator or denominator. For example, a decline in the total population of a county would lead to an increase in the share of seniors, even though the number of seniors does not change. In such a case, the relationship between e.g. deposits or loans and the change in the share of seniors would be driven by the decline in population.⁷ Detailed population data by age cohort are provided by the National Cancer Institute Surveillance, Epidemiology, and End Results (SEER) program. We use these data also to construct changes in the size of other age cohorts.

Bank exposure. To calculate banks' exposure to aging counties, we use data from the Federal Deposit Insurance Corporation's (FDIC) Summary of Deposits (SOD), which provides information on the geographic distribution of bank deposits.⁸ We compute

⁷In regressions of the log of the share of seniors on the log of the total number of seniors (the numerator) and the log of the total population (the denominator), a Shapley decomposition of the R-squared shows that around two-thirds of the variation in the share is explained by the variation in the total population, and around one-third by the variation in the senior population. Using the share of seniors as an explanatory variable would hence lead to significant measurement issues, as most of the variation would be due to changes in the total population. Instead, we directly control for population growth in our regressions.

⁸An assumption underlying our analysis is that deposits raised by banks in a given county are raised from residents of that county. Amel et al. (2008) show that between 1992 and 2004, the median distance between a depositor and its bank was three miles and remained constant over time. Nonetheless, the rise of online banking could mean that "branchless" deposits are recorded in the bank's headquarters branch. To the extent that the amount of branchless deposits is uncorrelated with bank exposure, such measurement error would arguably not lead to a bias in the estimates. In line with this argument, we find that excluding banks' headquarter county from the analysis does not affect our results (unreported).

banks' beginning-of-sample exposure as

$$exposure_b = \sum_{c} \frac{deposits_{b,c}}{deposits_b} \times \Delta old_c, \tag{1}$$

where $deposits_{b,c}$ and $deposits_b$ denote bank b's deposits in county c and its total deposits as of 1997. Δold_c is county c's change in the log of the population of ages 65 and above. High $exposure_b$ implies that a large share of banks' initial deposits is held in aging counties, while low exposure implies that deposits are held in counties with a small increase in the number of seniors. Higher exposure thus corresponds to an increase in deposit-weighted average aging in banks' borrower counties. Exposure is constructed from beginning-of-sample deposit shares, alleviating concerns about banks selectively opening branches in aging counties.⁹

Instrumental variable strategy. Counties that experience a stronger increase in seniors could differ along other dimensions that could matter for deposit growth. We thus predict Δold , i.e., the change in the population of ages 65 and above from 1997 to 2007, with the change in the population of age 45 to 65 from 1977 to 1987 in the same county. In essence, we use the predetermined component of each county's age structure – 20 years prior to our sample period – as an instrumental variable for the actual change in the age structure. This approach builds on the assumption that the historical demographic structure is plausibly exogenous to *changes* in contemporaneous confounding factors. For example, it purges Δold from changes in life expectancy, in- or out-migration, or incomes over the sample period.

Similar to county aging, bank exposure could be correlated with other (unobservable) factors that affect bank behaviour. For example, aging counties could, for whatever reason, face better economic prospects in terms of their unemployment trajectory or income growth. Banks more exposed to aging counties would then also be more exposed to faster growing areas, which poses a threat for our identification strategy. To address this concern beyond the fixed effects strategy (explained below), we construct exposure in Equation (1) based on Δold predicted by the historical demographic structure. In addition, and to test whether exposure to aging counties is correlated with exposure to counties with better growth trajectories, we compute exposure to the changes in local

⁹In the Online Appendix, we show that neither local aging nor bank exposure predict in which counties banks open new branches prior to 1997, see Table OA3.

unemployment rates and income per capita, analogous to Equation (1). We obtain a strong first stage, with an F-statistic of almost 60 (see also Figure OA2).

LTI ratios, loan and deposit growth. Home Mortgage Disclosure Act data provide detailed information on banks' residential mortgage lending.¹⁰ HMDA covers the vast majority of applications and approved mortgage loans in the U.S. The data include the application outcome (granted or denied), loan amount, and borrower income for each loan.¹¹ We measure bank risk-taking through the loan-to-income (LTI) ratio, defined as loan volume over applicant income. The LTI ratio is a significant predictor of ex-post default (Fuster et al., 2021) and has been widely used in the literature to measure the riskiness of loans (Dell'Ariccia and Marquez, 2006; Dell'Ariccia et al., 2012; Duchin and Sosyura, 2014). We compute the change in the average LTI ratio, as well as at the 10^{th} , 25^{th} , 50^{th} , 75^{th} , and 90^{th} percentile in each bank-county cell:

$$\Delta LTI_{b,c} = \frac{LTI_{b,c}^{07} - LTI_{b,c}^{97}}{(LTI_{b,c}^{07} + LTI_{b,c}^{97})/2}.$$
(2)

Additionally, we compute the change in the share of denied applications in each bankcounty cell. For these measures of risk-taking, we are restricted to the 'intensive margin', in the sense that we can only take into account counties in which banks made loans in 1997 and 2007.

Further, we compute the change in loan amounts and deposits at the bank-county level as

$$\Delta y_{b,c} = \frac{y_{b,c}^{07} - y_{b,c}^{97}}{(y_{b,c}^{07} + y_{b,c}^{97})/2},\tag{3}$$

where y is either HMDA loans or deposits. For our mortgage analysis, we use mortgage loans that were not sold in the respective calendar year. Since these loans are mostly retained on banks' balance sheets, they are predominately funded by deposits (Han et al., 2015; Cortés and Strahan, 2017). To account for entry into and exit out of counties over the long time horizon, we standardize the change in variables by their respective mid-

¹⁰We follow the literature and restrict the sample to conventional or Federal Housing Administration (FHA)-insured loans, exclude multi-family properties, and keep only originated, approved, and purchased loans. We also drop all observations with missing county Federal Information Processing Standards (FIPS) codes or missing borrower income, as well as loans extended to borrowers residing outside of metropolitan statistical areas (MSAs).

 $^{^{11}}$ In 2007, mortgage lending averaged around 30% of banks' total lending, and 40% for the largest banks. Additionally, HMDA data represent the most detailed publicly available data on bank lending disaggregated at the geographical level, which is why we focus on mortgage lending in our analysis.

points. This definition bounds growth rates to lie in the interval [-2, 2], where -2 implies that a bank exited a county between 1997 and 2007, and 2 that it entered.¹² Finally, we define the dummy *no branch* that takes on a value of one if bank *b* had no branch in county *c* in 1997 and zero otherwise.

Bank and county data. The FDIC provides detailed bank balance sheet data in its Statistics on Depository Institutions (SDI). We collect 1997 second quarter data on banks' total assets, Tier 1 capital ratio, nonperforming loans (NPL), return on assets, total deposits, total liabilities, share of non-interest out of total income, and overhead costs (efficiency ratio). We also include an indicator of an institution's primary specialization in terms of asset concentration that takes on ten distinct values.¹³ We also collect 1997 and 2007 data on banks' total deposits, total liabilities, and total loans and compute the change in the logarithm of each variable.

At the county level, we further collect public information on debt-to-income ratios (Federal Reserve Bank of New York Consumer Credit Panel, available from 1999 onward). We also collect 1997 data on the log of the population (NCI SEER), the unemployment rate (Bureau of Labor Statistics, Local Area Unemployment Statistics (BLS LAUS)), log income per capita (Bureau of Economic Analysis, Local Area Personal Income (BEA LAPI)), house price indices (Federal Housing Finance Agency (FHFA)), as well as employment shares in manufacturing (SIC code 20), retail trade (SIC code 52), and services (SIC code 70), provided in the County Business Patterns (CBP). CBP also provide information on employment in tradable and nontradable industries. Following Adelino et al. (2015), we classify two-digit NAICS code 23 as construction; codes 44, 45, and 72 as nontradable, and all others as tradable industries.

To remove outliers, we winsorize all variables at the 0.5^{th} and 99.5^{th} percentile. We then trim all remaining extreme values that lie at least five standard deviations above or below the mean.

¹²While the log difference is symmetric around zero, it is unbounded above and below, and does not easily afford an integrated treatment of entry and exit. The growth rate used in this paper is divided by the simple average in t-1 and t. It is symmetric around zero, lies in the closed interval [-2,2], facilitates an integrated treatment of entry and exit, and is identical to the log difference up to a second order Taylor series expansion (Davis and Haltiwanger, 1999).

¹³The indicator reflects, for example, whether a bank has an agricultural or a mortgage lending specialization.

Survey of Consumer Finances. The Survey of Consumer Finances (SCF) of the Federal Reserve provides detailed information on the allocation of households' financial assets. The SCF is a triennial cross-sectional survey on household assets and demographics. We combine the survey waves from 1998 and 2007 (41,366 observations). We collect information on respondents' total financial assets, deposits, and debt, as well as dummy indicators for whether they borrowed in the past year, had any outstanding debt, or whether they have been turned down for credit, or feared being denied credit in the past 5 years. As control variables, we further use data on the education level, number of kids, gender, race, marital status, home ownership, and a dummy for business ownership.

2.2 Descriptive statistics

In the average county, the number of seniors (Δold_c) increased by 12%,¹⁴ with a standard deviation of 15%. These numbers suggest significant variation in population aging across U.S. counties.

Table 1 provides descriptive statistics for the main variables. In total, our sample includes 1,843 banks for which we have data on loans and deposits over the sample period. Panel (a) summarizes bank *exposure* and other balance sheet characteristics from the SDI. For the average bank, exposure equals 0.10, with a standard deviation of 0.11. As exposure reflects deposit-weighted aging, a mean of 0.10 implies that the number of seniors increased by 10% in the average county where a bank takes deposits.¹⁵

Panel (b) reports summary statistics at the bank-county level. The average bank saw little change in its LTI ratio, but there is sizeable dispersion in the change across banks, the standard deviation equals 85 pp. While LTI ratios at the 25^{th} percentile declined by 19 pp, LTI ratios increased by more than 5 pp on average at the 75^{th} percentile, and by more than 20 pp at the 90^{th} percentile. The share of denied loans increased by 7 pp on average. For the average bank-county pair, deposits increased by 114% over the time period (13,086 observations); mortgage lending (one- to four-family residences) increased by 103% (53,197 observations).¹⁶ Along the intensive margin (disregarding bank entry and exit across counties), lending increased by 84% (17,643 observations).¹⁷

 $^{^{14}{\}rm The}$ unweighted county average of 11.6% in the change in the number of seniors compares to a US-wide increase in the senior population of around 10% over the sample period.

¹⁵Figure OA1 shows the distribution of aging and exposure to aging.

¹⁶The difference in the number of observations reflects that the average bank lends to counties in which it does not raise deposits.

 $^{^{17}}$ The unweighted average increase in mortgage lending in our sample is lower than the figure based

Finally, panel (c) reports summary statistics for county-level variables for the set of 2,163 counties.

Balancedness. To examine the balancedness in beginning-of-sample covariates among our sample of banks, in Table 2 we split banks into those below (low exposure) and above (high exposure) the median of the exposure distribution. High-exposure banks are slightly smaller (the difference in size is significant at the 10% level). They are statistically similar in terms of the share of nonperforming loans, return on assets, capital ratio, ratio of deposits to liabilities, or efficiency ratio (reflecting overhead costs).¹⁸ There is also no significant difference in the share of C&I loans or the share of loans extended to no-branch counties.

A potential concern for identification is that banks strategically opened branches to benefit from future deposit inflows in anticipation of demographic trends. In the Online Appendix, we show that neither local aging nor bank exposure predict in which counties banks open new branches prior to 1997 (see Table OA3). Specifically, we regress different measures that indicate whether banks opened branches on population aging or bank exposure and find no systematic relationship. These results mitigate concerns about such selection effects.

3 Population aging and bank risk-taking

The unprecedented rise in the senior population has led to a debate on whether population aging leads to a build-up of risks in the financial sector. High savings rates among seniors might lead to an abundance of savings and depressed returns, thereby potentially encouraging banks to reach for yield and engage in risk-taking (IMF, 2019). Higher risktaking by financial institutions in turn could give rise to financial instabilities (Brookings,

on aggregate outstanding mortgage debt (one- to four-family residences) that increased by around 190% from 1997 to 2007. See Federal Reserve Bank of St. Louis, FRED series MDOTP1T4FR.

¹⁸A potential concern arising from the correlation between bank size and exposure is that the estimates of exposure on bank lending and risk-taking reflect (unobservable) bank characteristics correlated with bank size. To address this concern, we estimate regressions with granular bank size*county fixed effects, where bank size corresponds to deciles of 1997 total bank assets. We thus account for any unobservable factors, including loan demand, that could differ across counties *and* be specific to banks of different sizes. In other words, we compare the effect of bank exposure to aging counties on lending and risk-taking among banks of near-identical size active in the same county. None of our results are affected by the inclusion of bank size*county fixed effects.

2019). In this section, we first analyze the effect of population aging on banks' loan-toincome ratios and non-performing loans. In the next section, we explore the underlying mechanism that drives the link between population aging and risk-taking.

3.1 Population aging and lending standards

To investigate the effects of an aging population on bank risk-taking, we estimate the following regression at the bank-county level:

$$\Delta LTI_{b,c} = \beta \Delta exposure_b + controls_{b/c} + \theta_c + \epsilon_{b,c}, \tag{4}$$

where $\Delta LTI_{b,c}$ is the change in bank b's average LTI ratio in mortgage lending in county c over the sample period. Variable exposure_b is bank exposure to aging counties at the beginning of the sample, as defined in Equation 1. Bank controls include log total assets, return on assets, nonperforming loans, total deposits over total liabilities, Tier-1 capital ratio, non-interest income, the indicator on bank specialization, and overhead costs, all at the beginning of the sample period. County controls include the change in the log of the total population, as well as the unemployment rate, log income per capita, the share of blacks, the log of the population, and the employment shares in manufacturing, retail, and services, also all at the beginning of the sample period. Standard errors are clustered at the bank and county level to account for correlations across observations among the same bank or county.

Equation (4) faces the common identification challenge to account for differences in the underlying characteristics of banks' borrowers. If more exposed banks lend to counties with different characteristics than less exposed banks, any observed differential change in LTI ratios reflects both county (demand) and bank (supply) factors. To address this challenge, we include granular borrower-county fixed effects (θ_c) that absorb any unobservable county characteristics, for example, changes in consumption, employment, or loan demand. Under the assumption that loan demand in a county is similar across banks, these fixed effects difference out demand forces and allow for identification of supply effects. Intuitively, these fixed effects enable us to compare two banks with different exposure that lend to the same county (Khwaja and Mian, 2008; Jiménez et al., 2014).

Table 3 reports the regression results for Equation (4) and shows that an increase in exposure to aging counties leads to higher LTI ratios. Column (1) shows that exposure

has a positive effect on bank risk-taking, significant at the 1% level. When we add bank controls in column (2), the coefficient of exposure remains similar in terms of sign, size, and significance. In line with results in Table 2, this finding suggests that exposure is not systematically correlated with other observable bank characteristics. Column (3) accounts for observable differences across borrower-counties by adding county controls, and column (4) further tightens identification by including borrower-county fixed effects. Holding all observable and unobservable county characteristics constant, exposure to aging counties still is associated with a highly significant increase in LTI ratios. The magnitude of the coefficient remains similar to column (2). Banks with a one standard deviation (sd) higher exposure see a relative increase in their LTI ratio by $(0.11 \times 0.42 =)$ 4.6 pp.

Including bank controls, county controls or county fixed effects in columns (2)-(4) does not materially change the size or significance of our coefficients relative to column (1), despite a substantial increase in the R². Our results hence suggest that the correlation between bank exposure and observable bank or unobservable county characteristics is low, reducing potential concerns about self-selection and omitted variable bias (Altonji et al., 2005; Oster, 2019).

To support the argument that exposure to aging counties leads to an increase in risktaking, column (5) replicates the specification in column (4), but uses bank exposure predicted with exposure constructed from the historical age structure (see Section 2). Supporting the notion that bank exposure is likely orthogonal to several bank and county characteristics, the coefficient on exposure remains positive and significant at the 1% level.¹⁹ These findings suggest a causal effect of bank exposure on LTI ratios.

Providing further support for a causal link from exposure to risk-taking, column (6) additionally controls for banks' exposure to the *changes* in counties' income per capita and unemployment rate. As discussed above, to the extent that aging counties might systematically differ in their economic prospects and growth trajectories, these exposure variables could explain changes in risk-taking and be correlated with exposure to aging counties. Accounting for these factors, column (6) shows that the coefficient on *exposure* declines only slightly in magnitude, but remains significant at the 1% level, which supports

¹⁹The first-stage F-statistic equals 57.97. The coefficient in the IV regression is larger than that in the OLS regressions, which could suggest that the IV overcomes measurement error in the exposure variable. Such mismeasurement could arise if, for example, seniors hold deposits in bank branches outside their residence county (which happens when e.g. snowbirds permanently move from the Great Lakes region to Florida but leave their deposits in a bank branch up north in the town of origin).

the plausible exogeneity of our IV.²⁰

Does the effect of exposure on risk-taking differ in counties where banks have a branch or not? In column (7), we interact exposure with the dummy *no branch*, which takes on a value of one in counties where banks had no branch at the beginning of the sample period. Results show that banks increase their LTI ratios by significantly more in counties where they have no branch. Column (8) confirms this finding once we add bank fixed effects to control for any unobservable bank-level characteristics. Previous work has shown that a greater borrower distance to the nearest branch requires banks to rely more on hard information (DeYoung et al., 2008). A greater distance can hence result in a less efficient screening and monitoring of borrowers, leading to the build-up of risks (Granja et al., 2018). These considerations suggest that the increase in LTI ratios in counties where banks have no physical presence could lead to material increases in credit risk.²¹ We will investigate this aspect in what follows.

3.2 Do laxer lending standards result in higher credit risk and losses?

The analysis in the previous section has shown that banks with higher exposure to aging counties increase their LTI ratios, which suggests an increase in credit risk. Did the increase in credit risk have consequences for the performance of exposed banks during a negative shock, and hence for financial stability more broadly?

Evidence from the Great Recessions suggests so: Figure 1, panel (a) shows the evolution of the ratio of nonperforming 1–4 family residential loans to assets from 2000 to 2015 for banks in the bottom (black dashed line) and top tercile (blue solid line) of the *exposure* distribution. Panel (b) depicts the same plot for overall nonperforming loans as a share of assets. Up until 2007, the series for high- and low-exposure banks are indistinguishable. After 2007 NPLs increase to a larger extent for high-exposure banks, relative

²⁰Formally testing for the sensitivity of the exclusion restriction following Conley et al. (2012) shows that even if the exclusion restriction would be substantially violated – i.e., the past demographic structure has a large and significant effect on the outcome variables during the sample period through channels other than contemporaneous aging – exposure leads to an economically sizeable and statistically significant increase in LTI ratios (unreported).

²¹Banks could have opened branches between 1997 and 2007 in counties where they had no branch in 1997, possibly mitigating informational frictions in the lending process. Consistent with this interpretation, we show that the coefficient on the interaction term *exposure* \times *no branch* increases in size when the dummy *no branch* takes on a value of one in counties where banks had no branches in 1997 *and* opened no branches until 2007 (see Table OA5).

to low-exposure banks. The difference across series peaks in 2010 and only reverts to similar levels by 2015. Figure 1 thus suggests that high exposure banks fared worse during the Great Recession, relative to low exposure banks. However, there was no discernible difference in pre-trends, i.e., greater risk-taking did manifest itself in higher NPL ratios only once the economy entered a downturn.

To investigate the effects of exposure on nonperforming loans in more detail, we estimate bank-level regressions of the following form:

$$\Delta y_b = \beta \ exposure_b + \delta \ exposure \ to \ \Delta hpi_b^{97-07} + controls_b + \epsilon_b, \tag{5}$$

where y_b is the 2007 to 2010 change in the ratio of nonperforming residential mortgages loans (1–4 family homes) to total assets or the ratio of overall nonperforming loans to total assets. Exposure denotes bank exposure to aging counties, predicted by the historical demographic structure. Bank controls include the log of total assets, return on assets, nonperforming loans, total deposits over total liabilities, Tier-1 capital ratio, non-interest income, the indicator on bank specialization, and overhead costs, all as of 1997. To ensure that our findings do not reflect that exposed banks are also more exposed to the pre-crisis housing boom, we construct banks' exposure to the change in county-level house price indices ('hpi') from 1997 to 2007 (*exposure to* Δhpi_b) analogously to Equation 1. In bank-level regressions, we can no longer control for possible confounding factors at the county level, so it is important to keep in mind that the coefficient β could at least in part reflect unobservable demand factors.

Column (1) of Table 4 shows that exposed banks see a significant increase in their NPL ratio on residential mortgage loans. Controlling for banks' exposure to the housing boom in column (2) does not affect the coefficient on exposure in any statistically or economically meaningful way. Columns (3) and (4) repeat the exercise, but use the change in banks' total NPL ratio as the dependent variable. Similar to columns (1) and (2), there is a strong positive and highly significant relationship: exposed banks see a stronger rise in nonperforming loans. In column (4), a one standard deviation increase in exposure is associated with an almost 13% stronger growth in nonperforming loans during the Great Recession (around 18% of the average growth in NPLs). While columns (1) and (2) directly follow from our main finding – exposed banks increase their provision of mortgage loans and increase their LTIs – findings in columns (3) and (4) suggest that risk-taking in mortgage lending had repercussions on overall bank performance during the crisis.

In conclusion, results in Table 3 suggest that banks exposed to aging counties increase their loan-to-income ratios, likely taking on credit risk. The effects are more pronounced in counties where banks have no physical presence – implying geographic spillover effects of population aging on financial stability. Table 4 shows that the aging-induced increase in risk-taking before the Great Recession led to a sharper increase in nonperforming loans during the crisis. Exposure to aging counties could therefore have exacerbated the negative effects of the Great Financial Crisis on bank health and financial stability – irrespective of banks' exposure to the housing boom.

In what follows, we explore the underlying mechanisms that drives the link between population aging and bank risk-taking.

4 Inspecting the mechanism: deposits and local loan demand

As we show in this section, banks operating in aging counties are subject to two opposing forces that affect their risk-taking behaviour. First, population aging leads to an increase in banks' available funding in the form of deposits. Second, an aging population reduces the local demand for credit.

4.1 Aging and deposits

Seniors have a higher propensity to hold deposits (Becker, 2007), consistent with a lower portfolio share of risky assets among older households (Fagereng et al., 2017). Panel (a) of Figure 2 plots average financial assets (left axis) and average deposits (right axis) for different age cohorts. Seniors, defined as the population of age 65 and above, hold on average about twice as much deposits as those in the cohort of 55-64, and more than three times as much as younger cohorts. Table OA2 in the Online Appendix shows that the positive relationship between age and deposits is not explained by cohort fixed effects, nor by a large set of individual-level controls such as income, occupation, homeownership or education.

Such differences in deposit holdings across age cohorts imply potentially large effects on banks' funding conditions when economies age, as deposits are a safe and cheap source of funding (Kashyap et al., 2002; Hanson et al., 2015; Carletti et al., 2021).²² Moreover, deposits by customers that have a longer relationship with a bank have been shown to be more stable (Iyer and Puri, 2012; Iyer et al., 2016) – which is the case for senior citizens.²³ Deposit inflows due to exposure to aging counties could hence be an important factor in explaining banks' risk-taking behavior.

Building on demographic variation in savings behavior, we investigate the relationship between secular changes in local demographics and bank deposits in the following crosssectional regression at the bank-county level:

$$\Delta deposit_{b,c} = \beta \ \Delta old_c + control_{b/c} + \theta_b + \epsilon_{b,c},\tag{6}$$

where $\Delta deposits_{b,c}$ is the change in bank b's deposits in county c over the sample period. The explanatory variable Δold_c is the change in the log county population of ages 65 and above. Due to a possible correlation between local aging and other county-level factors, we instrument Δold_c with the historical demographic structure, as explained in Section 2. Note that we focus on the change in the number of seniors, rather than the share, to avoid that results are spuriously driven by changes in the total population (see the discussion in Section 2.1). Instead, we control for the change in the total population directly.²⁴ County controls include the unemployment rate, log income per capita, the share of blacks, the log of the population, the change in the total population as well as employment shares in manufacturing, retail, and services, all at the beginning of the sample period. Bank controls include the log of total assets, return on assets, nonperforming loans, total deposits over total liabilities, Tier-1 capital ratio, non-interest income, the indicator on bank specialization, and overhead costs, all at the beginning of the sample period. Standard errors are clustered at the bank and county level to account for correlations across observations among the same bank or county.

In Equation (6), a coefficient $\beta > 0$ would indicate that local aging leads to an increase in local bank deposits. The underlying intuition is that the elderly hold more deposits,

 $^{^{22}}$ In 1997 the average bank in our sample had a ratio of deposits to total liabilities of 94%. A large literature highlights the importance of deposits for bank lending (see e.g. Gilje et al. (2016)).

 $^{^{23}}$ According to the Survey of Consumer Finances, respondents of age 65 and above are almost twice as likely than younger cohorts to state that they chose their financial institution because they "banked there a long time".

 $^{^{24}}$ Consider how aging could affect bank deposits. While aging is expected to have a positive effect on deposits (see Becker (2007) and the discussion in Section 4), measuring aging by the change in the share of seniors can spuriously generate a negative influence, due to a decline in total population – the denominator of the share.

as shown in Figure 2. Yet, β reflects differences in households' savings behavior and in bank behavior, both of which could affect individuals' incentives to hold deposit. For example, banks could decide to offer more attractive rates on deposits. To isolate the effect of aging on deposits, in some specifications, we include bank fixed effects (θ_b) that absorb all unobservable bank characteristics. In essence, we compare the effect of aging in different counties on deposits of the same bank, as we hold all bank characteristics constant.

Table 5 shows that an increase in the number of seniors in a county increases bank deposits. Column (1) includes county controls and shows that aging has a positive effect on bank deposits, significant at the 1% level. Once we add bank controls in column (2), the coefficient is similar in terms of sign, size, and significance. While we cannot include depositor-county fixed effects in Equation (6), in column (3) we include depositor-state fixed effects to control for confounding factors at the state level. The coefficient on Δ old declines in magnitude, but remains positive and significant at the 1% level.

In column (4), we further tighten identification and include bank fixed effects. Holding all observable and unobservable bank characteristics constant, local aging still is associated with a highly significant increase in deposits. Banks operating in a county with a one standard deviation (sd) larger increase in the number of seniors see a relative increase in local deposits by $(0.15 \times 0.640 =) 9.6\%$ (or around 8.4% of the average growth in deposits over the sample period). These results suggest a causal effect of local aging on bank deposits. To further support this interpretation, column (5) additionally controls for the 1997–2007 change in counties' income per capita and unemployment rate (note that regressions already account for the change in the total population). In principle, differences in economic prospects could affect deposits and be correlated with population aging. Accounting for these factors, column (5) shows that the coefficient on Δold remains positive and significant at the 5% level.²⁵

4.2 The local demand for credit

Beyond the positive effect on deposits, an aging population could reduce the local demand for credit. Seniors usually own their property and have high savings, so they have less need to borrow. In addition, seniors are also significantly less likely to start new companies

 $^{^{25}}$ In unreported regressions we show that higher exposure increases banks' total deposits and total liabilities. This suggests that banks do not offset the increase in deposits with a reduction in other liabilities.

(Azoulay et al., 2020), further dampening the demand for credit.²⁶ Consistent with lower demand for credit by seniors, Figure 2, panel (b) shows that older cohorts have significantly lower debt levels than younger cohorts (left axis). For the cohort of age 65 and above, total debt is close to zero. The share of respondents stating that they 'currently do not borrow' (right axis) also reflects this: While around 5% of younger cohorts state that they did not borrow, almost every third respondent of age 65 and above reports that he or she did not borrow.

We confirm these differences in borrowing behavior with respondent (i)-level data from the SCF. We estimate the following regression:

$$y_i = \beta \ respondent \ age \ 65_i + controls_i + \tau_t + \epsilon_i.$$

The independent variable respondent age 65_i is a dummy with a value of one if the respondent is of age 65 and above, and zero otherwise. Table 6 shows results. In columns (1)-(2), the dependent variable is a dummy that takes on a value of one if a respondent indicated that he/she has not borrowed over the past year. Column (1) shows that seniors are 20.9% more likely not to have borrowed, compared to younger respondents. Adding an extensive set of household-level controls (the log of total financial wealth, the education level, number of kids, gender, race, marital status, home ownership, and a dummy for business ownership), as well as survey wave fixed effects, narrows the gap only slightly in column (2). We obtain similar results when we use a dummy that takes on a value of one if a respondent indicated that he/she has no outstanding household debt in columns (3)-(4). For example, column (4) suggests that seniors are around 36% less likely to have any debt, compared to younger respondents. In principle, infrequent borrowing and no outstanding debt could also result from supply factors, e.g. age-specific discrimination by financial institutions that leads to higher denial rates. To investigate this explanation, columns (5)-(6) use a dummy for whether a respondent indicated that he/she has been turned down for credit or feared being denied credit in the past 5 years. Results show that seniors are significantly less likely to have been turned down for credit, which is consistent with the argument that seniors' demand for credit is lower than that

²⁶Further, an increase in the number of seniors leads to a decline in the labor force participation rate. As firms' marginal product of capital increases in the amount of available labor, a fall in the supply of labor reduces firms' demand for capital (Auclert et al., 2021). Several studies argue that aging contributes to slower growth and reduces investment opportunities (Fernald and Li, 2019). Maestas et al. (2016) and Gagnon et al. (2016) provide state-level evidence for the U.S. that aging leads to lower growth, mainly due to declining labor productivity. Aksoy et al. (2019) show a negative effect of aging on growth for a sample of OECD countries.

of younger respondents.

To provide further evidence that an increase in the number of seniors reduces the *local* demand for credit, we decompose mortgage loan growth in each bank-county cell into supply and demand factors, following Amiti and Weinstein (2018). To do so, we start from the following equation: $\Delta HMDA = \alpha_c + \beta_b + \varepsilon_{c,b}$, where the dependent variable is the growth in the amount of all mortgages originated by bank *b* in county *c*. The borrower-county demand channel is captured by α_c , while β_b denotes the bank supply channel. As described in Amiti and Weinstein (2018), this equation implies an adding up constraint that makes it possible to recover α_c , and β_b through a weighted least squares regressions, with weights given by the beginning-of-period share of total loans of bank *b* originated in county *c* over total loans originated in county *c*. For our regressions, we aggregate demand factors to the county level and investigate how they are affected by local aging. The Online Appendix provides further details on the methodology.

Table 7 shows that higher growth in the number of seniors has a significant and economically sizeable negative effect on the credit demand factor. Column (1) shows that an increase in the number of seniors leads to a significant decline in the demand for mortgages. Adding a battery of county-level controls in column (2) confirms this finding. In column (3), we use the instrumental strategy that we explain in Section 2 and confirm that the local demand for credit declines as the number of seniors increases. When we further control for *changes* in contemporary county characteristics in column (4), results remain qualitatively unaffected.

4.3 The geography of lending

Taken together, the analysis in this section suggests that banks operating branches in aging counties are subject to deposit inflows, but at the same time face lower demand and fewer investment opportunities in these counties. These two forces could force banks to look for new clients, in particular in counties where they operate no branches. To investigate the effects of population aging on bank lending, we estimate Equation (4) with the change in banks' mortgage lending in each borrower county as the dependent variable. In each regression, we include baseline bank controls and county fixed effects, thereby absorbing any unobservable county characteristics.

Table 8 shows that more exposed banks have higher loan growth over the sample period. In column (1), we regress the change in mortgage loan amounts on exposure

and bank controls. The coefficient of interest is positive and significant at the 1% level. Column (2) shows that this positive relation remains when we include county controls. Once we further account for unobservable county characteristics in column (3) through county fixed effects, the coefficient remains almost identical in terms of size, sign, and significance to that in column (2). The economic magnitude is sizeable: a one standard deviation increase in exposure is associated with an increase in bank lending by $(0.11 \times 1.284 =) 14.1\%$.

How does banks' exposure to aging counties shape the geography of their lending? If aging lowers the local demand for credit while increasing deposit inflows, banks may channel their loan supply to counties where they have no branches and are not directly exposed to the aging-induced decline in credit demand. To investigate this possibility, columns (4) and (5) interact exposure with the dummy *no branch* that takes on a value of one in counties where a bank operates no branches. In column (4), we find that more-exposed banks' loan growth is significantly higher in counties where they have no branch. In terms of magnitude, the increase in lending is almost 50% stronger in no-branch counties. Once we add bank fixed effects in column (5) to control for any unobservable bank-level characteristics, the interaction term increases in magnitude and remains significant at the 1% level. The results in Table 8 thus suggest that banks' exposure to aging counties leads to a significant and economically sizeable increase in credit, especially in counties where banks have no branches – in line with the findings for loan-to-income ratios, which suggested that banks relax lending standards especially in counties where they have no branches.

Does exposure also increase banks' aggregate lending? Column (6) shows results from a bank-level regression of the change in the log of banks' total loans on bank exposure and bank-level control variables. Exposed banks see significantly higher growth also in their total lending. This suggests that the aging-induced increase in mortgage credit does not crowd out other forms of lending. Instead, banks' loan volume increases overall, consistent with an overall increase in banks' funding.

5 Additional tests and robustness

Table 9 investigates bank risk-taking in more depth. We first investigate the LTI ratio distribution in columns (1) and (2) by looking at the 10^{th} and 90^{th} percentile separately. We find that the impact of exposure is stronger in the right tail of the LTI ratio distri-

bution. Columns (3)–(4) further show that banks increase their LTI ratios by more in counties where they have no branch – and again especially in the right tail of the LTI ratio distribution. A large literature shows that banks with lower capital ratios engage more in risk-taking when their funding conditions ease (Altunbas et al., 2014; Jiménez et al., 2014), as cheaper access to funds relaxes banks' capital constraints arising from moral hazard problems (Adrian and Shin, 2010). Column (3) thus interacts exposure with banks' Tier-1 capital ratio. It shows that higher exposure leads to a significantly smaller increase in LTI ratios for well-capitalized banks.²⁷

The rest of the columns use the share of loans denied as the dependent variable. Column (6) shows that more-exposed banks also denied significantly fewer loans, and column (7) shows that this is more pronounced among weakly capitalized banks. Column (8) reveals that – similar to findings for the LTI ratio – exposed banks deny fewer loans in counties where they operate no branches. Taken together, results in Table 9 complement our main finding that exposure to aging counties increases banks' risk-taking: banks with higher exposure increase risk-taking, especially among borrowers with higher LTI ratios, and they deny fewer loans.

Online Appendix. In the Online Appendix, we provide further details on our instrumental variable strategy (see Figure OA2). We also establish that neither local aging nor bank exposure systematically predict in which counties banks open new branches prior to 1997, ameliorating concerns about banks' strategically opening branches to benefit from the aging-induced rise in savings (see Table OA3).

We further show that controlling for for changes in the prime working age population or young population, either directly or via banks' exposure does not affect our results (see Table OA4). These results suggest that the estimated coefficients on aging do not spuriously reflect a correlation between the growth in the senior population and population growth in other demographic groups. Further, including bank size*county fixed effects, thereby taking into account potential differences in demand factors across banks of different sizes, does not materially affect our estimates (see Table OA5).

Finally, we find that counties in which exposed banks have a larger market share see a stronger increase in household debt-to-income ratios (see Figure OA3). The increase

 $^{^{27}}$ The interaction term is insignificant for LTI ratios at the 10^{th} percentile, but significant at the 1% level at the 90^{th} percentile, suggesting that higher capital ratios reduce risk-taking especially in the riskier segments (unreported).

in risk-taking and *loan*-to-income ratios at the bank level is mirrored in an increase of households' *debt*-to-income ratios.

6 Conclusion

Exploiting geographic variation in the change in seniors across the U.S., we establish that banks more exposed to aging counties increase their loan-to-income ratios by more and experience a sharper rise in nonperforming loans during the Great Recession. Inspecting the mechanism, we find that an aging-induced inflow of deposits, together with a decline in the local demand for credit, explain the link between population aging and bank risktaking.

These findings help shed light on how a major macroeconomic trend – population aging – affects the financial sector through changes in the supply of and demand for capital. Our novel results have implications for policy: Population aging could lead to financial instability, and through geographic spillovers do so even in regions and countries that are not directly affected by an aging population. As advanced economies face an unprecedented increase in the number of seniors over the next decade, this is a worrying finding. However, the fact that risk-taking is more pronounced for banks with lower capital ratios could mean that prudent capital regulation could limit the negative consequences of population aging for financial stability.

References

- Adelino, Manuel, Antoinette Schoar, and Felipe Severino (2015) "House prices, collateral, and self-employment", *Journal of Financial Economics*, 117 (2), pp. 288–306.
- Adrian, Tobias and Hyun Song Shin (2010) "Financial intermediaries and monetary economics", *Handbook of monetary economics*, 3, pp. 601–650.
- Aksoy, Yunus, Henrique S. Basso, Ron P. Smith, and Tobias Grasl (2019) "Demographic structure and macroeconomic trends", American Economic Journal: Macroeconomics, 11 (1), pp. 193–222.
- Altonji, Joseph G., Todd E. Elder, and Christopher R. Taber (2005) "Selection on observed and unobserved variables: Assessing the effectiveness of Catholic schools", *Journal of Political Economy*, 113 (1), pp. 151–184.
- Altunbas, Yener, Leonardo Gambacorta, and David Marques-Ibanez (2014) "Does monetary policy affect bank risk?", *International Journal of Central Banking*, March.
- Amel, Dean F, Arthur B Kennickell, and Kevin B Moore (2008) "Banking market definition: Evidence from the Survey of Consumer Finances", *Working Paper*.
- Amiti, Mary and David E Weinstein (2018) "How much do idiosyncratic bank shocks affect investment? Evidence from matched bank-firm loan data", *Journal of Political Economy*, 126 (2), pp. 525–587.
- Auclert, Adrien, Hannes Malmberg, Frédéric Martenet, and Matthew Rognlie (2021) "Demographics, wealth, and global imbalances in the twenty-first century", *NBER Working Paper*, 29161.
- Azoulay, Pierre, Benjamin F Jones, J Daniel Kim, and Javier Miranda (2020) "Age and high-growth entrepreneurship", American Economic Review: Insights, 2 (1), pp. 65–82.
- Becker, Bo (2007) "Geographical segmentation of US capital markets", Journal of Financial Economics, 85, pp. 151–178.
- Berger, Allen N and Gregory F Udell (2004) "The institutional memory hypothesis and the procyclicality of bank lending behavior", *Journal of Financial Intermediation*, 13 (4), pp. 458–495.
- Brookings (2019) "Should we fear population aging?", Future Development Blog (May).
- Butler, Alex and Hanyi Yi (2021) "Aging and public financing costs: Evidence from u.s. municipal bond markets", *Journal of Public Economics*, 2011 (104665).
- Carletti, Elena, Filippo De Marco, Vasso Ioannidou, and Enrico Sette (2021) "Banks as patient lenders: Evidence from a tax reform", *Journal of Financial Economics*, 141 (1), pp. 6–26.
- Carvalho, Carlos, Andrea Ferrero, Fernanda Nechio et al. (2017) "Demographic transition and low US interest rates", *FRBSF Economic Letter*, 27.
- **CGFS** (2018) "Financial stability implications of a prolonged period of low interest rates", (July).
- Conesa, Juan Carlos and Timothy J Kehoe (2018) "An introduction to the macroeconomics of aging", Journal of the Economics of Ageing, 11, pp. 1–5.
- Conley, Timothy G, Christian B Hansen, and Peter E Rossi (2012) "Plausibly exogenous", *Review of Economics and Statistics*, 94 (1), pp. 260–272.
- Cortés, Kristle Romero and Philip E. Strahan (2017) "Tracing out capital flows: How

financially integrated banks respond to natural disasters", *Journal of Financial Economics*, 125 (1), pp. 182–199.

- Cravino, Javier, Andrei A Levchenko, and Marco Rojas (2022) "Population aging and structural transformation", *American Economic Journal: Macro*, 14 (4), pp. 479–498.
- Davis, Steven J. and John C. Haltiwanger (1999) "Gross job flows", Handbook of Labor Economics, 3 (B), pp. 2711–2805.
- De Jonghe, Olivier, Hans Dewachter, Klaas Mulier, Steven Ongena, and Glenn Schepens (2020) "Some borrowers are more equal than others: Bank funding shocks and credit reallocation", *Review of Finance*, 24 (1), pp. 1–43.
- Dell'Ariccia, Giovanni, Luc Laeven, and Gustavo A Suarez (2017) "Bank leverage and monetary policy's risk-taking channel: Evidence from the United States", *Journal of Finance*, 72 (2), pp. 613–654.
- Dell'Ariccia, Giovanni and Robert Marquez (2006) "Lending booms and lending standards", The Journal of Finance, 61 (5), pp. 2511–2546.
- Dell'Ariccia, Giovanni, Deniz Igan, and Luc Laeven (2012) "Credit booms and lending standards: Evidence from the subprime mortgage market", Journal of Money, Credit and Banking, 44 (2-3), pp. 367–384.
- **DeYoung, Robert, Dennis Glennon, and Peter Nigro** (2008) "Borrower-lender distance, credit scoring, and loan performance: Evidence from informational-opaque small business borrowers", *Journal of Financial Intermediation*, 17 (1), pp. 113–143.
- **Doerr, Sebastian, Thomas Drechsel, and Donggyu Lee** (2021) "Income inequality, financial intermediation, and small firms", *BIS Working Paper*, No 944.
- **Doerr, Sebastian and Philipp Schaz** (2021) "Geographic diversification and bank lending during crises", *Journal of Financial Economics*, 140 (3), pp. 768–788.
- **Drechsler, Itamar, Alexi Savov, and Philipp Schnabl** (2021) "Banking on deposits: Maturity transformation without interest rate risk", *The Journal of Finance*, 76 (3), pp. 1091–1143.
- **Duchin, Ran and Denis Sosyura** (2014) "Safer ratios, riskier portfolios: Banks' response to government aid", *Journal of Financial Economics*, 113 (1), pp. 1–28.
- **ECB** (2018) "The economic impact of population ageing and pension reforms", *ECB Economic Bulletin*, 2.
- Fagereng, Andreas, Charles Gottlieb, and Luigi Guiso (2017) "Asset market participation and portfolio choice over the life-cycle", *Journal of Finance*, 72 (2), pp. 705–750.
- Fernald, John and Huiyu Li (2019) "Is slow still the new normal for GDP growth?", FRBSF Economic Letter, 17.
- Fuster, Andreas, Matthew C Plosser, and James I Vickery (2021) "Does CFPB oversight crimp credit?", CEPR Discussion Paper.
- Gabaix, Xavier (2011) "The granular origins of aggregate fluctuations", *Econometrica*, 79 (3), pp. 733–772.
- Gagnon, Etienne, Benjamin K. Johannsen, and David Lopez-Salido (2016) "Understanding the new normal: The role of demographics", *Working Paper*.
- Gilje, Erik P., Elena Loutskina, and Philip E. Strahan (2016) "Exporting liquidity: Branch banking and financial integration", *Journal of Finance*, 71 (3), pp. 1159–1184.
- Granja, João, Christian Leuz, and Raghuram Rajan (2018) "Going the extra mile:

Distant lending and credit cycles", NBER Working Paper, No. 25196.

- Han, Joongho, Kwangwoo Park, and George Pennacchi (2015) "Corporate taxes and securitization", *The Journal of Finance*, 70 (3), pp. 1287–1321.
- Hanson, Samuel G., Andrei Shleifer, Jeremy C. Stein, and Robert W. Vishny (2015) "Banks as patient fixed-income investors", *Journal of Financial Economics*, 117 (3), pp. 449–469.
- Heider, Florian, Farzad Saidi, and Glenn Schepens (2021) "Banks and negative interest rates", Annual Review of Financial Economics, 13, pp. 201–218.
- **IMF** (2019) "Macroeconomics of aging and policy implications", *Group of 20 IMF Staff Note* (July).
- Ioannidou, Vasso, Steven Ongena, and José-Luis Peydró (2015) "Monetary policy, risktaking, and pricing: Evidence from a quasi-natural experiment", *Review of Finance*, 19 (1), pp. 95–144.
- Iyer, Rajkamal and Manju Puri (2012) "Understanding bank runs: The importance of depositor-bank relationships and networks", *American Economic Review*, 102 (4), pp. 1414– 45.
- Iyer, Rajkamal, Manju Puri, and Nicholas Ryan (2016) "A tale of two runs: Depositor responses to bank solvency risk", *The Journal of Finance*, 71 (6), pp. 2687–2726.
- Jiménez, Gabriel, Steven Ongena, José-Luis Peydró, and Jesús Saurina (2014) "Hazardous times for monetary policy: What do twenty-three million bank loans say about the effects of monetary policy on credit risk taking?", *Econometrica*, 82 (2), pp. 463–505.
- Justiniano, Alejandro, Giorgio E. Primiceri, and Andrea Tambalotti (2019) "Credit supply and the housing boom", *Journal of Political Economy*, 127 (3), pp. 1317–1350.
- Kashyap, Anil K., Raghuram G. Rajan, and Jeremy C. Stein (2002) "Banks as liquidity providers: An explanation for the coexistence of lending and deposit-taking", *Journal of Finance*, 62 (1), pp. 33–73.
- Khwaja, Asim I. and Atif Mian (2008) "Tracing the impact of bank liquidity shocks: Evidence from an emerging market", *American Economic Review*, 98 (4), pp. 1413–1442.
- Kopecky, Joseph and Alan M Taylor (2022) "The murder-suicide of the rentier: Population aging and the risk premium", *Working Paper*.
- Kundu, Shohini, Seongjin Park, and Nishant Vats (2022) "The geography of bank deposits and the origins of aggregate fluctuations", *Working Paper*.
- Leahy, John V, Aditi Thapar et al. (2022) "Age structure and the impact of monetary policy", American Economic Journal: Macroeconomics, 14 (4), pp. 136–173.
- Liberti, José María and Mitchell A Petersen (2019) "Information: Hard and soft", Review of Corporate Finance Studies, 8 (1), pp. 1–41.
- Lin, Leming (2020) "Bank deposits and the stock market", *Review of Financial Studies*, 33 (6), pp. 2622–2658.
- Maddaloni, Angela and José-Luis Peydró (2011) "Bank risk-taking, securitization, supervision, and low interest rates: Evidence from the Euro-area and the US lending standards", *Review of Financial Studies*, 24 (6), pp. 2121–2165.
- Maestas, Nicole, Kathleen J Mullen, and David Powell (2016) "The effect of population aging on economic growth, the labor force and productivity", *NBER Working Paper*, No. 22452.

- Mendoza, Enrique G and Marco E Terrones (2008) "An anatomy of credit booms: Evidence from macro aggregates and micro data", *NBER Working Paper*, No. 14049.
- **OECD** (2019) "Fiscal challenges and inclusive growth in ageing societies".
- **Oster, Emily** (2019) "Unobservable selection and coefficient stability: Theory and evidence", Journal of Business & Economic Statistics, 37 (2), pp. 187–204.
- **Rehbein, Oliver and Steven Ongena** (2021) "Flooded through the back door: The role of bank capital in local shock spillovers", *Journal of Financial and Quantitative Analysis*.
- Saiz, Albert (2010) "The geographic determinants of housing supply.", Quarterly Journal of Economics, 125 (3), pp. 1253–1296.
- Smolyansky, Michael (2019) "Policy externalities and banking integration", Journal of Financial Economics, 132 (3), pp. 118–139.

Figures and tables





(a) Nonperforming 1-4 family loans

Note: This figure shows the evolution of nonperforming 1-4 family residential loans (over total assets) in panel (a) and of total nonperforming loans (over total assets) in panel (b). We split the sample into banks that lie in the top (high exposure, blue solid line) and bottom (low exposure, black dashed line) tercile of the distribution of bank exposure (as defined in Equation 1). Each series presents the average across all banks in the respective tercile. Banks with high exposure see a stronger increase in nonperforming loans from 2007 to 2010, relative to banks with low exposure. There are no differential pre-trends in the period before 2007.

Figure 2: Assets and debt by age group



(a) Assets and deposits

Note: This figure uses data from the Survey of Consumer Finances (1995-2007). Panel (a) plots total financial assets in \$1,000 on the left axis and total deposits in \$1,000 on the right axis for the average household in each age bin. Panel (b) plots total debt in \$1,000 on the left axis and the fraction of respondents answering *yes* to the question of whether they borrowed money on the right axis for each age bin. Older households are wealthier and hold more deposits; they also have less debt and are less likely to borrow.

	mean	sd	min	max	count
Panel (a): Bank level					
exposure	0.10	0.11	-0.12	0.55	1843
$\log(assets)$	12.06	1.27	10.14	19.62	1843
non-performing loans (%)	1.69	4.20	-11.82	30.77	1843
return on assets $(\%)$	1.16	0.56	-2.85	4.05	1843
deposits (%)	94.10	7.64	37.43	99.95	1843
tier 1 capital ratio $(\%)$	17.87	9.53	7.69	84.74	1843
non-interest income (%)	79.96	81.75	1.90	617.53	1843
efficiency ratio	61.47	14.14	27.27	174.52	1843
Δ deposits	0.75	0.53	-0.41	3.21	1843
Δ liabilities	0.79	0.54	-0.38	3.22	1843
Δ loans	0.89	0.59	-0.70	3.46	1843
Δ NPL 2007-2010	0.72	1.17	-2.58	6.47	1664
Δ NPL 2007-2010 (1-4 family)	0.51	0.99	-1.98	5.55	1664
Panel (b): Bank-county level					
Δ loan-to-income (mean)	-0.04	0.85	-2.94	3.28	20979
Δ loan-to-income (p10)	-0.25	0.80	-3.17	2.83	20979
Δ loan-to-income (p25)	-0.19	0.81	-2.95	2.73	20979
Δ loan-to-income (p75)	0.05	1.04	-3.37	3.79	20979
Δ loan-to-income (p90)	0.20	1.32	-4.64	6.18	20979
Δ deposits	1.14	1.05	-2.00	2.00	13086
Δ hmda	1.03	1.45	-2.00	2.00	53197
Δ hmda (intensive)	0.84	1.76	-4.01	6.37	17643
Δ denied	0.07	0.23	-0.60	0.83	20979
no branch dummy	0.91	0.29	0.00	1.00	53197
Panel (c): County level					
Δ old	0.12	0.15	-0.34	1.01	2163
log(population)	10.75	1.18	7.82	16.04	2163
share black	0.09	0.14	0.00	0.85	2163
unemployment rate	0.05	0.03	0.01	0.28	2163
log(income p.c.)	9.96	0.20	9.07	11.15	2163
employment share manufacturing	0.23	0.14	0.00	0.84	2163
employment share retail trade	0.24	0.06	0.02	0.57	2163
employment share services	0.30	0.09	0.06	0.99	2163

Table 1: Descriptive statistics

Note: This table shows descriptive statistics (mean, standard deviation, minimum, maximum, and number of observations) for the main variables at the bank, bank-county, and county level. All variables in levels are as of 1997, a ' Δ ' denotes 1997 to 2007 changes, unless indicated otherwise. For variable definitions, see Section 2.

	low e	low exposure high		exposure	mean diff.
	mean	sd	mean	sd	
$\log(assets)$	12.11	(1.26)	12.01	(1.26)	0.10*
non-performing loans $(\%)$	1.55	(0.38)	1.83	(0.45)	-0.28
return on assets $(\%)$	1.10	(0.55)	1.22	(0.57)	-0.12
deposits $(\%)$	93.99	(7.41)	94.18	(7.94)	-0.19
tier 1 capital ratio (%)	18.80	(10.66)	16.97	(8.42)	1.83
efficiency (%)	61.67	(14.08)	61.28	(14.21)	0.39
share CI loans (%)	11.96	(12.08)	13.85	(11.31)	-1.89
no branch ($\%$ of counties)	77.04	(42.08)	80.24	(39.84)	-3.20
Observations	922		921		1843

Note: This table shows results for a balancedness test of 1997 bank covariates. Banks with low (high) exposure are defined as banks with exposure below (above) the median of the distribution of exposure (as defined in Equation 1). *mean* denotes the mean and *sd* the standard deviation, *mean diff.* denotes the differences in means. We test for the statistical significance of the difference in means by regressing the exposure dummy on control variables in a logistic regression. For variable definitions, see Section 2.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
					IV	IV	IV	IV
VARIABLES	Δ LTI (avg)							
exposure	0.443***	0.450***	0.461^{***}	0.420***	0.698^{***}	0.620***	0.226^{*}	
	(0.068)	(0.071)	(0.073)	(0.076)	(0.091)	(0.091)	(0.131)	
no branch							-0.121***	-0.180***
							(0.019)	(0.019)
exposure \times no branch							0.505^{***}	0.569^{***}
							(0.165)	(0.168)
Observations	20,500	20,500	20,500	20,500	20,500	20,500	20,500	20,108
R-squared	0.002	0.059	0.065	0.109				
Bank Controls	-	\checkmark	\checkmark	\checkmark	\checkmark	Δ	Δ	-
County Controls	-	-	\checkmark	-	-	-	-	-
County FE	-	-	-	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Bank FE	-	-	-	-	-	-	-	\checkmark

Table 3: Exposure to aging counties and LTI ratios

Note: This table shows results for Equation 4 at the bank-county level. The dependent variable is the change in the average loan-to-income (LTI) ratio. *exposure* denotes bank exposure to aging counties (as defined in Equation 1). *no branch* is a dummy that takes on a value of one in bank-county pairs in which a bank does not operate branches in 1997, and zero otherwise. Columns (6) and (7) include bank exposure to the 1997 to 2007 change in income per capita and the unemployment rate as additional bank-level controls. For fixed effects and controls, see table footer. Standard errors are clustered at the bank and county level. The Anderson-Rubin F-Statistic equals 57.97. For variable definitions, see Section 2. *** p < 0.01, ** p < 0.05, * p < 0.1.

	(1)	(2)	(3)	(4)
VARIABLES	Δ NPL (mort)	Δ NPL (mort)	Δ NPL	Δ NPL
exposure	0.874^{***}	0.860***	1.148^{***}	1.134^{***}
	(0.306)	(0.300)	(0.327)	(0.320)
exposure to Δ hpi 1997-07		0.296***		0.301***
		(0.088)		(0.084)
Observations	1,661	1,661	$1,\!661$	$1,\!661$
Bank Controls	\checkmark	\checkmark	\checkmark	\checkmark

Table 4: Nonperforming loans during the Great Recession

Note: This table shows results for regressions at the bank level. The dependent variable is the 2007 to 2010 change in the share of nonperforming residential mortgages loans (columns (1) and (2)); the 2007 to 2010 change in the share of nonperforming loans (columns (3) and (4)); and the 2007 to 2010 change in total lending, standardized by pre-crisis assets (columns (5) and (6)). *exposure* denotes bank exposure to aging counties (as defined in Equation 1), constructed from the predicted change in seniors. *exposure to* Δ *hpi 1997-07* denotes deposit-weighted bank exposure to the increase in county-level house prices from 1997 to 2007. Each regression includes bank controls as of 1997. The Anderson-Rubin F-Statistic equals 8.34. Standard errors are robust. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)
VARIABLES	Δ deposits				
Δ old	0.793***	0.763***	0.561^{***}	0.640***	0.471**
	(0.294)	(0.280)	(0.195)	(0.203)	(0.203)
Observations	$12,\!947$	$12,\!947$	$12,\!947$	12,942	12,244
County Controls	\checkmark	\checkmark	\checkmark	\checkmark	Δ
Bank Controls	-	\checkmark	\checkmark	-	-
State FE	-	-	\checkmark	\checkmark	\checkmark
Bank FE	-	-	-	\checkmark	\checkmark

Table 5: Population aging and local deposits

Note: This table shows results for Equation 6 at the bank-county level in columns (1)–(6). The dependent variable is the change in deposits. Δold denotes the change in the log of the county population of ages 65 and above, predicted with the change in the population of age 45 to 65 from 1977 to 1987 in each county. Column (6) includes the 1997 to 2007 change in income per capita and the unemployment rate as additional county-level controls. For fixed effects and controls, see table footer. Standard errors are clustered at the bank and county level. The Anderson-Rubin F-Statistic equals 16.38. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	never borrow	never borrow	has any debt	has any debt	turned down	turned down
respondent age 65+	0.209^{***} (0.003)	0.188^{***} (0.006)	-0.422*** (0.004)	-0.358*** (0.008)	-0.251^{***} (0.005)	-0.158^{***} (0.005)
Observations	41,366	41,366	41,366	41,366	41,366	41,366
R-squared	0.080	0.101	0.176	0.247	0.058	0.178
Survey wave FE	-	\checkmark	-	\checkmark	-	\checkmark
Controls	-	\checkmark	-	\checkmark	-	\checkmark

Table 6: Seniors are less likely to borrow

Note: This table shows results for the following regression $y_i = \beta$ respondent age $65_i + controls_i + \tau_t + \epsilon_i$, where the age group 17-64 is the omitted category. The dependent variable in columns (1)–(2) is a dummy variable that takes on a value of one if a respondent indicated that he/she has not borrower over the past year. The dependent variable in columns (3)–(4) is a dummy variable that takes on a value of one if a respondent indicated that he/she has no outstanding household debt. The dependent variable in columns (5)–(6) is a dummy variable that takes on a value of one if a respondent indicated that he/she has been turned down for credit or feared being denied credit in the past 5 years. Columns (2), (4), and (6) add an extensive set of household-level controls: the log of total financial wealth, the education level, number of kids, gender, race, marriage status, home ownership, and a dummy for business ownership. Source: Survey of Consumer Finances 1998 and 2007. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)
			IV	IV
VARIABLES	Δ loan demand (AW)			
Δ old	-0.563***	-0.733**	-0.796**	-0.968***
	(0.206)	(0.318)	(0.357)	(0.252)
Observations	754	754	754	745
R-squared	0.008	0.203		
County Controls	-	\checkmark	\checkmark	Δ

Table 7:	Population	aging	and	the	demand	for	credit
10010 1.	I opulation	~ <u>8</u> 8	ana	0110	aomana	101	orouro

Note: This table shows results for regressions at the county level. The dependent variable is the demand factor of a decomposition of HMDA loan growth (1997-2007) following Amiti and Weinstein (2018). The dependent variable is standardized to a mean of zero and standard deviation one. Δ old denotes the change in the county population of ages 65 and above. Column (3) instruments Δ old with the change in the population of age 45 to 65 from 1977 to 1987. Column (4) includes the 1997 to 2007 change in income per capita and the unemployment rate as additional county-level controls. Standard errors are robust. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)	(6)
						bank
VARIABLES	Δ hmda	Δ loans				
exposure	1.625***	1.383***	1.284***	0.723***		0.635***
	(0.093)	(0.094)	(0.092)	(0.184)		(0.161)
no branch				0.858***	0.653***	
				(0.023)	(0.023)	
exposure \times no branch				0.746^{***}	0.549***	
				(0.204)	(0.196)	
Observations	51,727	51,727	51,727	51,727	$51,\!659$	1,750
Bank Controls	\checkmark	\checkmark	\checkmark	\checkmark	-	\checkmark
County Controls	-	\checkmark	-	-	-	-
County FE	-	-	\checkmark	\checkmark	\checkmark	-
Bank FE	-	-	-	-	\checkmark	-

Table 8: Bank exposure and lending

Note: This table shows results for regressions at the bank-county level in columns (1)-(5) and at the bank level in column (6). The dependent variable is the change in bank-county HMDA loans that are not sold by the end of the year in columns (1)-(5); and the change in total loans in column (6). *exposure* denotes bank exposure to aging counties (as defined in Equation 1), constructed from the predicted change in seniors. *no branch* is a dummy with a value of one for bank-county pairs in which a bank does not operate branches in 1997, and zero otherwise. For fixed effects and controls, see table footer. Standard errors are clustered at the bank and county level in columns (1)-(5). Column (6) uses robust standard errors. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
VARIABLES	Δ LTI (p10)	Δ LTI (p90)	Δ LTI (p10)	Δ LTI (p90)	Δ LTI (avg)	Δ denied	Δ denied	Δ denied
exposure	0.298^{***}	0.948^{***}			1.337^{***}	-0.179^{***}	-0.684^{***}	
	(0.083)	(0.133)			(0.146)	(0.021)	(0.035)	
no branch			-0.247***	-0.129^{***}				0.027***
			(0.018)	(0.032)				(0.005)
exposure \times no branch			0.058	1.048***				-0.077**
			(0.149)	(0.266)				(0.036)
exposure \times Tier-1 capital ratio					-3.534***		2.599^{***}	
					(0.623)		(0.142)	
Observations	20,500	20,500	20,108	20,108	20,500	20,500	20,500	20,108
R-squared	0.092	0.102	0.301	0.346		0.158	0.173	0.464
Bank Controls	\checkmark	\checkmark	-	-	\checkmark	\checkmark	\checkmark	-
County FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Bank FE	-	-	\checkmark	\checkmark	-	-	-	\checkmark

Table 9: Robustness tests

Note: This table shows results for Equation 4 at the bank-county level. The dependent variable is the change in the loan-to-income (LTI) ratio at different percentiles in columns (1)–(5); and the change in the share of loans denied in columns (6)–(8). *exposure* denotes bank exposure to aging counties (as defined in Equation 1). The variable *no branch* is a dummy with a value of one for bank-county pairs in which a bank does not operate branches in 1997; *Tier-1 capital* denotes banks' Tier-1 capital ratio (as of 1997); Columns (3), (4), and (8) include bank and county fixed effects. The remaining columns include bank controls and county fixed effects. Standard errors are clustered at the bank and county level. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

A Online Appendix

Variable name	Description	Source
Bank level		
	bank exposure to aging counties (deposit-weighted)	EDIC SOD NCI SEEB
A deposite	change in total denosits	FDIC SDI
A loops	change in total hank loans	FDIC SDI
Δ mortgages	change in total residential mortgage loans	FDIC SDI
log(assets)	log total assats	FDIC SDI
non-performing loans (%)	share of NPL over total loans	FDIC SDI
BOA (%)	raturn on assats	FDIC SDI
denosits (%)	total denosits over total liabilities	FDIC SDI
tier 1 capital $(\%)$	tier 1 capital ratio	FDIC SDI
non-interest income $(\%)$	non-interest income over average assets	FDIC SDI
Δ NPL (mort)	change in net charge-offs on mortgage loans 2007-10	FDIC SDI
Δ NPL	change in net charge-offs on all loans 2007-10	FDIC SDI
A loans/asset	change in loans over pre-crisis assets 2007-10	FDIC SDI
	change in toalis over pre-crisis assets 2007-10	T DIO 5DI
Bank-county level		
$\overline{\Delta}$ deposits	Change in deposits	FDIC SOD
Δ HMDA	Change in mortgage loans	HMDA
Δ LTI (mean)	Change in average loan-to-income ratio	HMDA
Δ LTI (pX)	Change in X-percentile loan-to-income ratio	HMDA
Δ denied	Change in share of denied mortgage loans	HMDA
County level		
Δ old	change in population 65+	NCI SEER
log(unemployed 1940)	log total unemployed 1940	ICPSR
log(population)	log total population	NCI Seer
unemployment rate	unemployment rate	BLS LAUS
participation rate	labor force participation rate	BLS LAUS
log(income p.c.)	log income per capita	BEA LAPI
employment share manufacturing	employment share of manufacturing sector (SIC 20)	CBP
employment share retail trade	employment share of retail trade sector (SIC 50)	CBP
employment share services	employment share services sector (SIC 70)	CBP
Δ debt-to-inc	Change in debt-to-income ratio	FRBNY
presence of exposed banks	loan-weighted average across bank exposure of banks active in county	HMDA, FDIC SOD, NCI SEER
<u>Other variables</u>		
elasticity	MSA housing supply elasticity	Saiz (2010)

Table OA1: Variable Definitions

Note: This table reports variable definitions and sources. Changes (Δ) are from 1997 to 2007, all other variables are as of 1997 (unless indicated otherwise). For details see Section 2.



Figure OA1: County aging and bank exposure – distribution

Note: This figures shows the distributions of the county-level log change in the population of age 65 and above from 1997 to 2007 in panel (a) and bank exposure as defined in Equation 1 in panel (b).



Figure OA2: Instrumental variable strategy

Note: Panel (a) plots counties' actual and predicted change in seniors. Δold denotes the change in county population age 65 and above, and Δold (predicted) denotes the change in county population of age 45 to 65 from 1977 to 1987. Panel (b) plots banks' actual and predicted exposure, where predicted exposure uses Δold (predicted).

Figure OA3: Change in county-level debt-to-income ratios



Note: This figure shows the average change in the county-level debt-to-income ratio from 1997 to 2007. We split the sample into counties that lie in the top, middle, and bottom tercile of local presence of exposed banks. presence is computed as the average exposure of banks active in a county, weighted by each banks local HMDA loan volume ($presence_c = \sum_b \frac{l_{b,c}}{l_c} \times exposure_b$, where $l_{b,c}$ and l_c denote bank b's HMDA loans in county c and county c's total HMDA loans (both as of 1997)). Counties with higher values of presence have a higher share of loans extended by high-exposure banks.

	(1)	(2)	(3)	(4)
VARIABLES	$\log(deposits)$	$\log(deposits)$	$\log(deposits)$	$\log(deposits)$
age group 35-64	0.848***	0.842***	0.320***	-0.209***
	(0.022)	(0.022)	(0.024)	(0.017)
age group $65+$	1.658^{***}	1.656^{***}	1.312***	0.258***
	(0.026)	(0.026)	(0.042)	(0.030)
$\log(\text{financial wealth})$				0.641^{***}
				(0.003)
Observations	$58,\!078$	$58,\!078$	$58,\!078$	$58,\!078$
R-squared	0.065	0.066	0.308	0.630
Survey wave FE	-	\checkmark	\checkmark	\checkmark
Controls	-	-	\checkmark	\checkmark

Table OA2: The relation between age and deposits

Note: This table shows results for the following regression $log(deposits)_i = age \ group_i + controls_i + \tau_t + \epsilon_i$, where the age group 17-34 is the omitted category. Column (3) adds an extensive set of household-level controls: education level, number of kids, occupation, gender, race, marriage status, home ownership, and a dummy for business ownership. Column (4) further controls for the log of respondents' overall financial wealth. Source: Survey of Consumer Finances 1992, 1998, and 2007. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	Δ branches	open br	entry	Δ branches	open br	entry
Δ old	-1.002	0.125^{*}	0.096			
	(0.677)	(0.074)	(0.082)			
exposure				1.192	0.206	0.287^{*}
				(1.358)	(0.126)	(0.153)
Observations	$16,\!977$	$16,\!977$	$16,\!977$	17,026	17,026	17,026
R-squared	0.173	0.390	0.387	0.115	0.199	0.158
County Controls	\checkmark	\checkmark	\checkmark	-	-	-
Bank FE	\checkmark	\checkmark	\checkmark	-	-	-
Bank Controls	-	-	-	\checkmark	\checkmark	\checkmark
County FE	-	-	-	\checkmark	\checkmark	\checkmark

Table OA3: Did banks open branches between 1994 and 1997?

Note: This table shows results for regressions at the bank-county level. The dependent variables are the change in the number of branches (columns (1) and (4)), a dummy with value one if a bank opened a branch in a county (columns (2) and (5)), and a dummy with value one if a bank entered a county (columns (3) and (6)). Δold denotes the log change in county population age 65 and above. *exposure* denotes bank exposure to aging counties (see Equation 1). Standard errors are clustered at the bank and county level. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	Δ deposits	Δ deposits	Δ deposits	Δ hmda	Δ hmda	Δ hmda
Δ old	0.762^{***}	0.857^{***}	0.670***			
	(0.108)	(0.116)	(0.139)			
exposure				1.694^{***}	1.681***	1.778***
				(0.111)	(0.113)	(0.112)
Observations	13,086	13,086	$13,\!086$	47,004	47,004	47,004
R-squared	0.350	0.350	0.351	0.209	0.209	0.209
County Controls	\checkmark	\checkmark	\checkmark	-	-	-
Bank Controls	-	-	-	\checkmark	\checkmark	\checkmark
Bank FE	\checkmark	\checkmark	\checkmark	-	-	-
County FE	-	-	-	\checkmark	\checkmark	\checkmark
Δ pop	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Δ young	-	\checkmark	\checkmark	-	\checkmark	\checkmark
Δ prime age	-	-	\checkmark	-	-	\checkmark

Table OA4:	Growth	and	exposure –	other	${\rm demographic}$	groups

Note: This table shows results at the bank-county level for regression equation (6) with the change in deposits as dependent variable in columns (1)-(3); and for regression equation (3) with the change in HMDA loans as dependent variable in columns (4)-(6). old denotes the population 65 and above. Each column in columns (1)-(3) controls for population growth in a different cohort (pop, young, prime working age, corresponding to the total population, population age 29 and younger, and population age 25-44, respectively). Each column in columns (4)-(6) controls for bank exposure to each of these groups. The different exposure measure are constructed as defined in Equation 1. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
					$97~\mathrm{and}~07$	$97~\mathrm{and}~07$		
VARIABLES	Δ hmda	Δ hmda	Δ LTI (avg)	Δ LTI (avg)	Δ LTI (avg)	Δ LTI (avg)	Δ denied	Δ denied
exposure	1.121***		0.849^{***}		0.132		-0.232***	
	(0.077)		(0.096)		(0.115)		(0.025)	
no branch		0.634***		-0.211***	-0.138***	-0.162***		0.027***
		(0.026)		(0.023)	(0.019)	(0.019)		(0.005)
exposure \times no branch		0.512^{**}		0.670***	0.756***	0.657***		-0.073*
		(0.202)		(0.191)	(0.153)	(0.160)		(0.044)
Observations	49,781	49,633	19,140	18,644	20,500	20,108	19,140	18,644
Bank Controls	\checkmark	-	\checkmark	-	\checkmark	-	\checkmark	-
Bank size*County FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Bank FE	-	\checkmark	-	\checkmark	-	\checkmark	-	\checkmark

Table OA5: Bank size*county fixed effects, permanent branch sample, and denied loans

Note: This table shows results for regressions at the bank-county level. The dependent variable is the change in bank-county HMDA loans in columns (1)–(2); the change in the LTI ratio in columns (3)–(6); and the change in the share of denied loans in columns (7)–(8). *exposure* denotes bank exposure to aging counties (as defined in Equation 1). In columns (3)–(4), *no branch* is a dummy with a value of one for bank-county pairs in which a bank does not operate branches in 1997, and zero otherwise. In columns (5)–(6), *no branch* is a dummy with a value of one for bank-county pairs in which a bank had no branches in 1997 and did not open any branches during the sample period, and zero otherwise. For fixed effects and controls, see table footer. Standard errors are clustered at the bank and county level. For variable definitions, see Section 2. *** p<0.01, ** p<0.05, * p<0.1.

Amiti-Weinstein Decomposition

To provide evidence on the negative influence of aging on local mortgage loan demand, we use the method developed by Amiti and Weinstein (2018). To understand the intuition behind this method, consider the following fixed effects model

$$\Delta L_{cbt} = \alpha_{ct} + \beta_{bt} + \epsilon_{cbt}, \tag{OA1}$$

where the growth in loans of borrower c (a county in our case) obtained from a lender b at time t is regressed on borrower-time (α_{ct}) and lender-time (β_{bt}) fixed effects. α_{ct} captures the component in loan variation explained by borrower-level variation and β_{bt} captures the component explained by lender-level variation. Amiti and Weinstein (2018) illustrate that the empirical counterparts of α_{ct} and β_{bt} provide estimates for loan demand and loan supply channels, respectively, if appropriately weighted.

Amiti and Weinstein (2018) develop their method by modifying the model in OA1 in two ways. The first modification ensures that the estimated borrower and lender shocks aggregate up to exactly match total loan growth in the economy. They show that this adding up constraint is satisfied when one uses lagged loan amounts as weights. The second modification establishes that the method incorporates both the formation and termination of lending relationships. This modification is done by changing the normalization by dropping the first borrower and lender from the estimation.

Formally, these modifications enable Amiti and Weinstein (2018) to obtain lenderand borrower-level shocks by solving the following system of B + C equations up to a numeraire:

$$D_{bt}^{B} = \frac{\sum_{f} L_{cbt} - \sum_{f} L_{cb,t-1}}{\sum_{c} L_{cb,t-1}} = \hat{c}_{t} + \hat{\ddot{\beta}}_{bt} + \sum_{c} \phi_{cb,t-1} \hat{\ddot{\alpha}}_{ct}$$
(OA2)

$$D_{ct}^{C} = \frac{\sum_{b} L_{cbt} - \sum_{b} L_{cb,t-1}}{\sum_{b} L_{cb,t-1}} = \hat{c}_{t} + \hat{\ddot{\alpha}}_{ct} + \sum_{b} \theta_{cb,t-1} \hat{\ddot{\beta}}_{bt}$$
(OA3)

where D represents the growth of bank's total lending, or firm's total borrowing, and \hat{c}_t is a time fixed effect. $\phi_{cb,t-1} \equiv \frac{L_{cb,t-1}}{\sum_c L_{cb,t-1}}$ and $\theta_{cb,t-1} \equiv \frac{L_{cb,t-1}}{\sum_b L_{cb,t-1}}$ show the weight of each loan for the banks and counties, respectively. $\hat{\beta}_{bt}$ and $\hat{\alpha}_{ct}$ are modified forms of $\hat{\beta}_{bt}$ and $\hat{\alpha}_{ct}$ in OA1, where $\hat{\beta}_{bt} \equiv \beta_{bt} - \beta_{1t}$ and $\hat{\alpha}_{ct} \equiv \alpha_{ct} - \alpha_{1t}$. Note that β_{1t} and α_{1t} are the

fixed effects of the first lender and borrower, which are dropped from the estimation for normalization purposes. In these equations, $\ddot{\beta}_{bt}$ captures the lender supply shocks and $\ddot{\alpha}_{ct}$ the borrower demand shocks. In words, this method explains a lender's aggregate loan growth by lender-specific loan supply factors and a weighted average of changes in loan demand of its borrowers. Similarly, a borrower's aggregate loan growth is driven by its loan demand and a weighted average of loan supply factors of all its lenders.

After obtaining α_{ct} and β_{bt} , we can decompose the aggregate loan growth D_t into three components.

$$D_t = (\bar{A}_t + \bar{B}_t) + \mathbf{W}_{t-1}^{\mathbf{B}} \Phi_{t-1} \dot{\mathbf{A}}_t + \mathbf{W}_{t-1}^{\mathbf{B}} \dot{\mathbf{B}}_t$$
(OA4)

The first component, $(\bar{A}_t + \bar{B}_t)$, shows the common shocks on aggregate lending and measures what happens to the lending of the median borrower-lender pair. The second components are vectors, $\dot{\mathbf{A}}_t$ and $\dot{\mathbf{B}}_t$, that stack borrower- and lender-level shocks, α_{ct} and β_{bt} . These vectors show the granular shocks a là Gabaix (2011) and are expressed as deviations from \bar{A}_t and \bar{B}_t . These vectors measure the importance of granular shocks on aggregate lending. Φ_{t-1} is a weighting matrix and $\mathbf{W}_{t-1}^{\mathbf{B}}$ is the share of lender *l*'s loan volume out of total lending by all lenders in year *t*.

Previous volumes in this series

1049 November 2022	Crypto trading and Bitcoin prices: evidence from a new database of retail adoption	Raphael Auer, Giulio Cornelli, Sebastian Doerr, Jon Frost and Leonardo Gambacorta	
1048 November 2022	Is the Covid-19 pandemic fast-tracking automation in developing countries? Evidence from Colombia	Leonardo Bonilla, Luz A Flórez, Didier Hermida, Francisco Lasso, Leonardo Fabio Morales, Juan Jose Ospina and José Pulido	
1047 November 2022	What drives inflation? Disentangling demand and supply factors	Sandra Eickmeier and Boris Hofmann	
1046 November 2022	The case for convenience: how CBDC design choices impact monetary policy pass-through	Rodney Garratt, Jiaheng Yu and Haoxiang Zhu	
1045 October 2022	The term structure of carbon premia	Dora Xia and Omar Zulaica	
1044 October 2022	Gender diversity in bank boardrooms and green lending: evidence from euro area credit register data	Leonardo Gambacorta, Livia Pancotto, Alessio Reghezza and Martina Spaggiari	
1043 October 2022	The scarring effects of deep contractions	David Aikman, Mathias Drehmann, Mikael Juselius and Xiaochuan Xing	
1042 September 2022	Emerging market bond flows and exchange rate returns	Peter Hördahl and Giorgio Valente	
1041 September 2022	The impact of fintech lending on credit access for U.S. small businesses	Giulio Cornelli, Jon Frost, Leonardo Gambacorta and Julapa Jagtiani	
1040 September 2022	Quantifying the role of interest rates, the Dollar and Covid in oil prices	Emanuel Kohlscheen	
1039 September 2022	Cyber risk in central banking	Sebastian Doerr, Leonardo Gambacorta, Thomas Leach, Bertrand Legros and David Whyte	
1038 September 2022	Building portfolios of sovereign securities with decreasing carbon footprints	Gong Cheng, Eric Jondeau and Benôit Mojon	
1037 August 2022	Big Techs vs Banks	Leonardo Gambacorta, Fahad Khalil and Bruno M Parigi	
1036 August 2022	The scarring effects of deep contractions	David Aikman, Mathias Drehmann, Mikael Juselius and Xiaochuan Xing	

All volumes are available on our website www.bis.org.